Author's response to reviews

Title: Evaluation of drinking patterns and their impact on alcohol-related aggression: a national survey of adolescents behaviours

Authors:

Valeria Siciliano (valeria.siciliano@ifc.cnr.it)
Lorena Mezzasalma (lorena.mezzasalma@ifc.cnr.it)
Valentina Lorenzoni (valentina.lorenzoni@ifc.cnr.it)
Stefania Pieroni (stefania.pieroni@ifc.cnr.it)
Sabrina Molinaro (sabrina.molinaro@ifc.cnr.it)

Version: 2 Date: 22 April 2013

Author's response to reviews: see over
To the Executive Editor of BMC Public Health
Natalie Pafitis
BMC-series Journals
BioMed Central
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HL

Dear Editor,

we would like to submit the revised version of the manuscript MS: 2757554039614938 with required changes.

Please note:
1. Copy-editing: the English language was edited.
2. Data availability: in the Methods section of the manuscript we have added information on data source and availability.

Hoping to hearing from you soon,

Sincerely yours,

Sabrina Molinaro, PhD
Institute of Clinical Physiology, Italian National Research Council (IFC-CNR)
Via Moruzzi 1
56124 Pisa
Italy
Phone: +39 050 315 20 94
Email: sabrina.molinaro@ifc.cnr.it